Author’s response to reviews

Title: Association of PM2.5 concentration with health center outpatient visits for respiratory diseases of children under 5 years old in Lima, Peru

Authors:

Jennifer Dávila Córdova (jdavila.1@alumni.unav.es)
Vilma Tapia (VILMA.TAPIA.A@upch.pe)
Bertha Vasquez (vasquezapestegui@gmail.com)
Luis Ordoñez (lordonez@dge.gob.pe)
Bryan Vu (bryan.vu@emory.edu)
Kyle Steenland (nsteenl@emory.edu)
Gustavo Gonzales (gustavo.gonzales@upch.pe)

Version: 2 Date: 29 Nov 2019

Author’s response to reviews:

Good Morning

It was made the change of the title of the article "Effects of PM2.5 concentration in health center outpatient visits for respiratory diseases of children under 5 years old in Lima, Peru: Towards new environmental policies" by "Association of PM2.5 concentration with health center outpatient visits for respiratory diseases of children under 5 years old in Lima, Peru". Also changes in title pages and conclusions.